

Open access • Journal Article • DOI:10.1177/0027950110364103

Managing housing bubbles in regional economies under emu: ireland and spain — Source link 🖸

Thomas Conefrey, John Fitz Gerald

Institutions: Economic and Social Research Institute

Published on: 01 Jan 2010 - <u>National Institute Economic Review</u> (SAGE PublicationsSage UK: London, England) Topics: Fiscal policy

Related papers:

- Resolving Ireland?s banking crisis
- Ireland in EMU: More Shocks, Less Insulation?
- This Time Is Different: Eight Centuries of Financial Folly
- · Assessing the role of income and interest rates in determining house prices
- The US and Irish Credit Crises: Their Distinctive Differences and Common Features



ECONSTOR

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Conefrey, Thomas; Gerald, John Fitz

Working Paper Managing housing bubbles in regional economies under EMU: Ireland and Spain

ESRI Working Paper, No. 315

Provided in Cooperation with: The Economic and Social Research Institute (ESRI), Dublin

Suggested Citation: Conefrey, Thomas; Gerald, John Fitz (2009) : Managing housing bubbles in regional economies under EMU: Ireland and Spain, ESRI Working Paper, No. 315, The Economic and Social Research Institute (ESRI), Dublin

This Version is available at: http://hdl.handle.net/10419/50134

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.

Mitglied der Libhiz-Gemeinschaft

WWW.ECONSTOR.EU



Working Paper No. 315

September 2009

Managing Housing Bubbles in Regional Economies under EMU: Ireland and Spain

Thomas Conefrey and John Fitz Gerald

Abstract: With the advent of EMU monetary policy can no longer be used to prevent housing market bubbles in regional economies such as Ireland or Spain. However, fiscal policy can and should be used to achieve the same effect. This paper shows that the advent of EMU relaxed existing financial constraints in Ireland and Spain, allowing a more rapid expansion of the housing stock in those countries to meet their specific demographic circumstances. However, the failure to prevent these booms turning into bubbles did lasting damage to the two economies, damage that could have been avoided by more appropriate fiscal policy action.

Corresponding Author: John.FitzGerald@esri.ie

ESRI working papers represent un-refereed work-in-progress by members who are solely responsible for the content and any views expressed therein. Any comments on these papers will be welcome and should be sent to the author(s) by email. Papers may be downloaded for personal use only.

Managing Housing Bubbles in Regional Economies under EMU: Ireland and Spain

1. Introduction¹

Over the last thirty years the housing market has been a source of significant economic instability on at least one occasion in each of a wide range of OECD economies. When housing bubbles have burst they have caused substantial damage to the affected economy. As a result, they ought to be a cause for concern for economic policy makers. On some occasions when housing bubbles have burst they have also been associated with significant disruption to the domestic financial system (e.g. in the mid-west of the US in the mid 1980s, in Scandinavia in 1989/1990 and today in the US, Spain and Ireland). Any such financial sector disruption greatly magnifies the macro-economic disruption which is caused by a housing sector collapse (Honohan and Klingebiel, 2003).

The analysis carried out in Girouard *et al.* (2006), indicates that when housing bubbles have occurred in the past they were essentially idiosyncratic in nature – they generally did not occur simultaneously across major regions of the OECD. The analysis in Himmelberg *et al.* (2005) also indicates that prior to the recent crisis there has not been a "US housing bubble" but rather a series of local bubbles in individual cities or regions in the US at different times over the last quarter of a century.

Until the current crisis there was little evidence that housing bubbles had become any more frequent over time and it is also interesting that, even with monetary union in the US, there had not, until now, been any obvious tendency for a "convergence" of bubbles within the US, much less in the EU. However, Girouard *et al.* (2006) suggest that global factors were important in continuing the housing boom across quite a number of markets in the period up to 2006 and this may explain why today there are

¹ This paper was prepared as part of an IRCHSS-funded project "Turning Globalization to National Advantage: Economic Policy Lessons from Ireland's Experience". It has benefited from comments from colleagues, especially Ide Kearney and David Duffy, from Philip Lane of TCD, as well as participants at the June 2009 EUROFRAME conference in London and participants at a meeting of National Economic Research Organisations, OECD, Paris, September, 2009.

housing market crises in an unusually large number of EU members as well as in the US.

This idiosyncratic pattern reflects the fact that the drivers of housing prices are varied in nature and they include significant local or regional variables. Himmelberg *et al.* (2005) refer to the role of "superstar cities" where local circumstances can result in a prolonged period of higher than average growth in house prices.

The evidence suggests that until recently housing markets in the Euro area mirrored the pattern in the US, with some national markets showing very rapid inflation in asset values and others showing no inflation at all. While there is today some synchronisation across certain EMU members in their housing cycles, there is still very considerable diversity across the EMU. For example, in Ireland and Spain there is currently a major down turn in the housing market while in Germany and Finland there is no evidence of any major reduction in housing prices or output.

While housing market crashes in individual economies have generally not had global implications they have had very severe repercussions domestically. This is the case today in Ireland and Spain. It is important to understand what drives these housing markets so that future bubbles might be avoided. Because of the serious damage that housing crashes can cause to individual economies it will be important in the future to develop policy instruments to manage housing markets and to prevent dangerous bubbles occurring. While monetary policy is not a suitable instrument within EMU for this purpose because of the idiosyncratic nature of housing markets, this paper argues that fiscal policy instruments can achieve the desired impact if used in an innovative fashion.

In Section 2 of this paper the demographic factors that underpin long-term demand for housing are considered with particular reference to Ireland and Spain. This paper goes on in Section 3 to set out a simple model of the housing markets in Ireland and Spain. In the context of these models Section 4 considers how the advent of EMU has changed the cost of capital facing households and how this has affected the experience of Ireland and Spain in the current decade. Section 5 discusses the significance of the building sector in these two economies, an important factor in determining the macroeconomic impact of a housing price crash. Section 6 discusses the implications of this experience for how housing markets should be managed in the future in EMU. Monetary policy, as implemented by the ECB, cannot be targeted at preventing dangerous situations arising in any individual regional housing market. The problem that this poses for members of EMU is discussed, together with the appropriate fiscal strategy for managing potential regional bubbles. Section 7 presents conclusions.

2. The Drivers of the Housing Market – Demographics

The desired stock of housing in the long run in any economy is a function of permanent income, the user cost of housing, the cost of building, the availability of credit, local factors, and, crucially, the demographic structure. Because houses typically have very long lives and because demographic circumstances in individual economies change slowly the new build each year is typically a small proportion of the actual stock of dwellings. While the legacy effects of the destruction caused by the Second World War required major investment in the 1950s and 1960s, in many European countries this reconstruction work is long since completed. However, there may be issues for the new EU 12 member states related to the legacy effects of a half a century under communist rule. Today there are still significant differences in demography across the EU-15 which imply there will be regional differences in the demand for housing in the long-term.

	Total dwellings	Occupied dwellings
Denmark	621	595
Estonia	599	521
France	634	526
Germany	599	
Hungary	519	475
Ireland		
2002	525	464
Ireland		
2006	574	478
Netherlands		534
Poland	454	421
Portugal	634	448
Spain	655	444
UK	575	551

Table 1:Dwellings per 1000 adults, 2001

Source: EUROSTAT file cens_rdhh for dwellings and EUROSTAT population file for population. Data for Ireland for 2006 are from CSO, *Census 2006*

Girouard *et al.* (2006) and ECB (2006) analyse the role of "fundamentals" in determining house prices. This research, and the research on individual EU economies (e.g. Duffy, 2002; Murphy, 2006 on Ireland, and Muellbauer and Murphy, 2008 for

the UK, Gonzalez and Ortega, 2009 for Spain), all indicate that while interest rates do affect house price inflation, it is only one of a range of driving variables. In the longer-term income growth and demographic change are very important drivers of changes in relative prices.

An important variable affecting the demand for new dwellings, and hence the investment in housing, is the existing stock of dwellings relative to the population. Table 1 shows the data for a number of EU countries for 2001, (and for Ireland for 2006) immediately after EMU began. For most of the EU 15 countries the stock of dwellings was high relative to the adult population (aged 20 and over) and had not changed much over the previous decade. This suggested that in terms of demographic factors these countries were close to their long run equilibrium stock. However, for Ireland, Portugal and Spain, when the ratio is calculated only for occupied houses, the stock is substantially lower relative to the adult population than for many of the other EU-15.²





Note: Ireland, 2006, Census of Ireland, Germany, 1991 and UK 1992, Labour Force Survey.

If preferences for the standard of domestic accommodation are similar across EU populations this would suggest that, because of the fact that Ireland and Spain have only recently converged to the EU average in terms of income per head, their endowment of housing infrastructure may still be well below what might be expected for countries with their current standard of living. Given the slow adjustment of the

² For Spain the stock of "unoccupied" dwellings is very high because of the importance of foreign investment in dwellings purely for holiday purposes.

housing stock, it would take many years of above average investment in housing before the stocks in those countries will have reached the current levels in countries such as France, Germany, the Netherlands and the UK.

A measure of the difference in endowment is the percentage increase in the stock of dwellings that would be needed to raise the number of dwellings per adult in Ireland, Spain, and Portugal to the number in, for instance, France. Using this measure an increase in the stock of dwellings of roughly an eighth in each of these three countries would be needed to bring the endowment to French levels. For some of the newer EU member states the gap is much larger; for Poland the necessary increase would amount to almost a quarter of the existing housing stock.

This difference in the endowment of housing infrastructure is also reflected in a difference in the proportion of people at each age group in Ireland who are heads of households relative to other EU countries, such as Germany and the UK. Figure 1 shows data for Ireland, Germany and the UK. For each five-year cohort it shows the proportion of the cohort who are heads of households.³ Obviously cultural differences, in terms of such factors as divorce etc., may affect the numbers of households. However, these data provide a useful indicator of the availability of dwellings. In spite of the relatively similar standard of living in the three countries the lower endowment of dwellings in Ireland means that adults have to share dwellings with their parents or unrelated friends under circumstances where adults in Germany or the UK would rent (or buy) their own apartment or house. This difference is particularly noticeable for those aged between 25 and 34. Garcia and Hernandez (2008) report similar circumstances in Spain where high prices have affected headship rates. They report that because of the high prices "This places an obstacle to the departure of young adults from the family home, delaying their emancipation age...".

The very rapid rise in the cost of housing over the last decade in Ireland has meant that the headship rates did not change at a time when living standards, measured in terms of real disposable income, rose dramatically. This may also have been the case for Spain. Now that house prices (and rents) are falling it may be anticipated that individuals who are currently sharing accommodation will, at some time in the future,

³ Technically referred to as household reference person. In the case of Germany if there is a male in the household he is normally defined as the reference person. In Ireland and the UK it is nominated by the household and an increasing share of them are now women.

enter the housing market seeking to establish a new independent household. ⁴ This overhang of potential renters or buyers will mean that, after the current adjustment, investment in housing is likely to remain above the average for the EU 15 for some time to come in Ireland, Portugal and Spain.



Figure 2: Average Annual Demand for Housing in Ireland

Figure 2 shows a decomposition of average annual housing demand in Ireland since 1991 together with forecasts for the future illustrating the impact of demographic factors. At unchanged headship rates around 20,000 additional dwellings a year are needed to take account of the rising number of adults, especially in the younger age groups.⁵ Because of the rapid rise in the price of housing the headship rates, which are well below those in Germany and the UK, remained relatively unchanged over the 1990s. This reflected the rapid rise in the cost of accommodation which offset the effects of higher incomes. In addition, immigration has accounted for approximately 15,000 of the additional dwellings occupied each year between 2003 and 2006. This reflected the fact that many immigrants to Ireland are skilled (well

Source: Bergin, et al., 2009

⁴ However, there is evidence that in spite of the very serious decline in the economy in 2008 the number of households showed a record increase in that year suggesting a rapid response of headship rates to a very big reduction in rents (Bergin *et al.*, 2009).

⁵ With a very small share of the population aged over 75 the numbers dying each year leaving dwellings vacant for the younger generation is low. In countries such as Germany the population is closer to demographic balance.

educated) and they have similar headship rates to natives (FitzGerald *et al.*, 2008). However, a very substantial share of the additional dwellings built between 2003 and 2006 were not occupied by permanent households. While some were built as holiday homes many of them were built for purely speculative purposes and the build up of the stock of vacant dwellings after 2003 presaged future problems in the housing market.

Both Spain and Ireland have a very small private rented sector with home ownership being the norm. In the case of Spain the home ownership rate in 2000-1 was 81 per cent.⁶ For Ireland it was around 75 per cent. The private rented sector was around 11 per cent in Spain and around 15 per cent in Ireland. This contrasts with home ownership rates in the UK and the US of just under 70 per cent and just over 40 per cent in Germany. This means that the private rental sector in both Ireland and Spain is very small and, as discussed later, this may have implications for the way the current adjustment to falling prices plays out.⁷

Spain is in a similar position to Ireland for dwellings that are permanently occupied. However, in the case of Spain what makes the position very uncertain is that a high proportion of the new build in recent years has been designed to satisfy demand from abroad for holiday homes. Some of these holiday homes are owned directly by foreigners and some are rented by Spanish owners (individuals or companies) to foreigners. While holiday homes are also a factor in the demand for housing in Ireland they have constituted a much smaller share of the demand than in the case of Spain. In the case of Spain the number of holiday homes or dwellings otherwise vacant is almost a third of the total stock whereas in Ireland it is around 15 per cent, still a relatively high number.

For holiday homes a range of additional factors are important. There are cultural effects, where changing preferences mean that people in Northern Europe favour holidays in the sun. In addition, the dramatic fall in the cost of travel has made frequent flights to Southern Europe affordable for many in countries such as Germany, the UK and Ireland.

⁶ Data from Garcia and Hernandez, 2008 and CSO: Census of Ireland, 2006.

⁷ According to the Irish consumer price index in April 2009 rents were down 13 per cent on April 2008.

Obviously, when a Dutch or a Swedish person chooses to buy an apartment in Spain it is their income rather than Spanish incomes that will be important. In addition, they have the option of funding the investment through local finance or through finance in their home country. However, in determining the user cost of such an investment it is the expectations as to property prices in Spain that matter rather than expectations of capital gains (or losses) in their home country.

Because holiday homes are a "luxury" it is to be expected that demand will be more volatile than in the case of demand for dwellings to live in. However, unless global warming changes the climate in the North Atlantic very rapidly, it is to be expected that there will be continuing long-term demand for new housing in coastal regions of the Mediterranean. There remains uncertainty about the phasing of that investment over time and the location of new development round the Mediterranean.

The data in Table 1 indicate that Poland has an even lower stock of dwellings per thousand adults than is the case for Spain and Ireland. In addition, in the case of Poland a lot of the stock of pre 1990 dwellings have major problems, in particular in terms of insulation. If Poland behaves in a similar manner to the converging EU-15 countries (Ireland, Greece, Spain and Portugal) there is likely to be significant demand for new housing in coming decades. While current low income levels relative to the EU average mean that priority is given to other forms of investment today, in the future rising incomes could see a major growth in demand for new housing. As in Spain and Ireland, managing these pressures in the future to ensure that the rest of the economy is not crowded out will be important.

3. A Model of House Prices

In this section we specify and estimate new house price equations for both Ireland and Spain. In both cases the equation for new house prices is based on the standard textbook model and consists of an inverted demand equation. This specification of the Irish and Spanish housing models draws on the work of Murphy (1998). In the textbook model house prices are determined by real disposable income, the per capita housing stock, a proxy for the user cost of housing (mortgage interest rate less the change in house prices) and the percentage of the population in the key household formation age group of 25-34 years. Other empirical studies including those by Kenny

(1999) and Roche (2001) have used similar specifications. The baseline equation for new house prices may be written as follows:

$$Log\left(\frac{PHNEW}{PC}\right)_{t} = \beta_{0} + \beta_{1}\log\left(\frac{Y}{POP}\right) - \beta_{3}\log\left(1 + \frac{HSTOCK}{POP}\right) + \beta_{4}\left(\frac{POP\ 2534}{POP}\right) - \beta_{5}(rr)$$
(1)

The standard model does not include any variables capturing the effects of financial liberalisation on house prices. Roche (2003) and McQuinn (2004) use alternative versions of the standard model by including an average mortgage loan variable as a proxy for financial liberalisation. However Murphy (2006) has criticised the inclusion of the average mortgage variable as an explanatory variable on a number of grounds including the argument that the mortgage loan variable is not independent of house prices. In the absence of a reliable measure of financial liberalisation, we follow the approach of Murphy (2006) and include a dummy for 2003 which is designed to pick up "the combined effects of financial liberalisation, policy interventions since 1998 and speculative frenzy effects".

In Equation (2) we apply the model to Irish data. Real new house prices (defined as new house prices (*PHNEW*) deflated by the personal consumption deflator (*PC*)) are modelled as a function of real disposable income per capita (*Y*), the housing stock (*HSTOCK*) per head of the total population (*POP*_i), the percentage of the population aged 25-34 years (*POP2534*) and the real cost of capital for housing (*rr*). A dummy variable is included for 2003 (*D03*)⁸. The user cost variable is calculated as the mortgage interest rate less the three period moving average of the rate of inflation in new houses⁹.

⁹ Real cost of capital for housing =
$$rr = \left[nom.int.rate_{t-1} - \frac{1}{2} \log(\frac{phnew_{t-1}}{phnew_{t-3}}) \right]$$

⁸ The dummy variable is set equal to zero up to and including 2003 and is set equal to one thereafter.

$$Log\left(\frac{PHNEW}{PC}\right)_{t} = 8.41 + 0.99 \log\left(\frac{Y}{POP}\right) - 3.79 \log\left(1 + \frac{HSTOCK}{POP}\right)$$
(2)
+ 14.12 $\left(\frac{POP \ 2534}{POP}\right) - 0.97(rr) + 0.11(D03)$

The full estimation results are shown in Appendix 1. The estimated coefficients on the equation for new house prices show the expected signs. The coefficient on income measures the responsiveness of house prices to changes in income. The elasticity of .9 is high and suggests a high degree of pass through of changes in income to new house prices. The coefficient on the per capita housing stock variable is significant and negatively signed indicating the importance of a scarcity effect. This implies that Ireland's relatively low endowment of dwellings coupled with strong population growth in the 1990s was an important driver of house prices. The demographic variable is positive and significant indicating that the expansion of the key house buying population cohort contributed to the increase in new house prices. The coefficient on the real cost of capital is significant and correctly signed. The coefficient on the 2003 dummy is significant and may be picking up the combined effects of a more liberalised mortgage market, demand and supply side policy measures which drove up house prices and the speculative frenzy which accompanied the rapid appreciation in the value of new houses.

We next apply the standard housing model with some adjustments to the Spanish data and draw comparisons with the estimation results for Ireland. In equation (3) new house prices are a function of real household disposable income per capita (Y), the housing stock (*HSTOCK*) per head of the total population (*POPt*), the percentage of the population aged 25-34 years (*POP2534*) and the real cost of capital for housing (*rr*). This mirrors the specification of Irish house prices in equation (2) and the standard house price model represented by equation (1). In addition two dummy variables are included for the years 1987 and 2003. The 1987 dummy is designed to capture the effect of Spain's accession to the EU which involved major liberalization in the Spanish economy, including changes in the financial system, while the 2003 dummy is included to pick up the bubble effect on house prices of the boom in the Spanish property market during the early 2000's. The full estimation results are shown in Appendix 1.

$$Log\left(\frac{PHNEW}{PC}\right)_{t} = -6.71 + 1.13 \log\left(\frac{Y}{POP}\right) - 8.22 \log\left(\frac{1 + HSTOCK}{POP}\right)$$
(3)
+ 25.61 $\left(\frac{POP \ 2534}{POP}\right) - 0.85 \log\left(rr\right) + 0.11(D\ 87) + 0.27 D(03)$

The estimated coefficients for new house prices in the equation for Spain show the expected signs, however the magnitude of the coefficients differ compared to results for Ireland. The estimation results for Spain show that a 1 per cent rise in real income per capita leads to a 1.1 per cent rise in house prices compared to a 0.9 per cent rise in Ireland. The housing stock per capita variable has a substantially bigger effect in the Spanish equation than in the Irish equation.

The estimation results for both Ireland and Spain show that the increase in the proportion of the population in the key household formation age group was an important driver of new house prices. A 0.01 unit rise in the household formation ratio leads to a 14.1 per cent rise in new house prices in Ireland and a 25.6 per cent rise in Spain. Lower interest rates also contributed to higher new house prices in both Ireland and Spain. The estimation results show that a one percentage point rise in the user cost variable leads to almost a 1 per cent fall in house prices in Ireland and a .9 per cent fall in Spain.

Faust et al (2001) have estimated a standard monetary policy reaction function for the Bundesbank based on a simple Taylor rule and have compared the ECB's actual monetary policy with the predictions from this reaction function, using Euro Area data. They find that ECB interest rates have been consistently below those predicted by the estimated Taylor rule. For Ireland Faust et al find that in 1999 had interest rates been determined by a Taylor rule the fitted target rate in that year would have averaged 9.2 per cent. This corresponds to a discrepancy of 6.4 per cent between the average predicted target rate and the actual rate that prevailed in that year. Applying this to the coefficient on the real interest rate in equation (2) this implies that had interest rates in Ireland been 6.4 per cent higher in 1999 this would have reduced house prices by around 6.2 per cent. For Spain Faust et al estimate that interest rates in 1999 would have been 4.7 per cent higher than the actual rate, using a Taylor rule. With an estimated coefficient of .85 on the real interest rate term in equation (3) this implies that house prices in Spain would have been reduced by a further 4 per cent had interest rates averaged the 7.3 per cent predicted by the Taylor rule rather than the average actual rate of 2.8 per cent which prevailed at the time. This is a simple measure of the initial impact of EMU membership on the housing markets in Spain and Ireland.

Finally two time dummies are included in the Spanish model. The coefficient on the 1987 dummy indicates that house prices in Spain in that year were 10 per cent higher after taking account of the other variables in the model. The positive and significant coefficient on the 1987 dummy variable picks up the impact of Spain's accession to the EU in 1986.

As in the equation for Ireland the 2003 dummy is included in the Spanish equation to capture the combined effects of financial liberalisation, policy interventions and speculative frenzy effects which gave rise to strong growth in Spanish house prices post 2003 – behaviour that cannot be captured in a standard model of house prices. The effect of the 2003 dummy is to raise house prices by about 27 per cent above the level explained by the other variables in the model. Without a measure of financial liberalisation it is difficult to establish the precise effect of more favourable financial conditions on house prices in Spain. However, as for Ireland, the Spanish housing boom was undoubtedly propagated by a combination of a more liberalised mortgage market, policy changes and frenzy activity as reflected in the significance and magnitude of the dummy variables in both equation (2) and (3).

Overall the estimation results for Ireland and Spain indicate that fundamentals such as income, demographics, the size of the existing housing stock and the user cost of capital are crucial determinants of new house prices. In addition, the results suggest that other factors such as financial liberalisation and frenzy effects are likely to have raised the level of house prices above where they otherwise would have been, particularly in the period after 2003 when both countries experienced a housing boom.

4. The Cost of Capital

Prior to EMU there were substantial differences in the cost of capital facing householders across the EU. These differences in the cost of capital may have slowed investment in housing in some countries where they were relatively high. However, the advent of EMU substantially changed the expected cost of capital facing agents in many member states of the Euro area. It meant that there was substantial convergence in the nominal interest rate facing similar agents across the EMU. However, because

of continuing differences in the rate of inflation across the EMU real interest rates still differ for households.

EMU "liberalised" credit markets

The financial sector has undergone major changes over the last thirty years. This has involved both liberalisation and the changes consequent on the move to Economic and Monetary Union. These two developments have combined to provide a dramatically different environment today for households wanting to finance investment in housing compared to thirty years ago. In the 1970s many households in Ireland and Spain were credit constrained and were not able to finance their optimal level of investment in housing. However, it was not just the availability of finance that has changed but its cost has also been reduced by the move to monetary union and also by financial liberalisation in the 1990s and the early years of this decade. Worldwide lax monetary policy over the last decade reduced the cost of capital, not just in Ireland and Spain, but also for households in most developed economies. However, the advent of EMU as well as financial liberalisation potentially reduced the cost of finance by making available a much wider pool of savings without incurring exchange rate risk.

In the Irish case, up to the end of the 1970s all mortgage finance was provided by "building societies". These financial institutions raised all their finance from deposits, generally by households. They were all mutual companies facing rather different incentives than private limited companies (that can be assumed to be profit maximisers). As a result, interest rates were not used as an instrument to clear the mortgage market and there was widespread credit rationing. In some cases the building societies operated queuing based on the length of time that a borrower had previously been a depositor.

Over the course of the 1980s the standard banking system moved into the mortgage market and building societies began to raise some of their funds from the interbank market rather than from small depositors. Also the building societies themselves moved away from the mutual model so that they had an incentive to use price (the interest rate) to clear the market.

There are a number of implications of these institutional changes. In the 1970s credit rationing was the norm in finance for housing. The interest rate was dependent on the availability of domestic savings and would have been little affected by exchange rate risks. The extent of rationing would have depended on the competition

for personal deposits and the demand for mortgages. However, over the course of the 1980s as the market liberalised and as financial institutions had access to the interbank market, the constraint on finance was relaxed. Instead the cost of finance was increasingly related to the cost of funds on the local interbank market.

	1980-1998	1985-1998
Ireland	2.0	1.6
Spain	1.4	1.9
$\rm UK^{10}$	2.1	

Table 2: Excess Returns from Borrowing in DMs and lending in local currency

The Impact of EMU – nominal interest rates

One of the key long-term reasons why membership of EMU was considered desirable for Ireland and many other potential members was the expected impact on interest rates and the cost of capital (Baker, Fitz Gerald and Honohan, 1996, Calmfors, 1997). For countries such as Spain, Ireland, and the UK, historically there had been a significant risk premium attaching to borrowing in their own currency relative to borrowing in DMs. Table 2 shows a measure of the risk premium, the excess returns for governments from borrowing in DMs and lending in the local currency, for two time periods, 1980-1999 and 1985-1999. This is calculated using medium-term government bonds – it is the risk premium that faced the government sector. The advent of EMU saw such differentials between yields on government bonds almost disappear. However, over the past year the differentials have again reappeared in a dramatic form as a consequence of the current financial crisis and the differing exposure of different countries to its consequences.

Thus for most of the member governments, as expected, the advent of EMU substantially reduced the cost of capital. This had a very striking positive impact on the public finances for countries that were heavily indebted, especially where the debt was financed on a short to medium term basis – Italy. The notable exception to this was Germany which had experienced a lower real cost of capital for some considerable time.

¹⁰ For the UK it is 1979-1998.

Multinational firms trading on the world market always had the opportunity to borrow in DMs, availing of the lower cost of finance. With diversified trade such firms could also hedge their currency risk within the firm itself, availing of the lowest available cost of finance internationally. In recent years, with financial liberalisation, many commercial firms in the EU also had access to financial instruments which allowed them to hedge some of their foreign currency risk at a price. This meant that for multinational firms the cost of capital was probably lower than for firms trading purely on the domestic market. This would suggest that the benefits of a reduction in nominal interest rates as a result of EMU, while significant for the company sector, were probably greatest for smaller firms.

For much of the private sector borrowing in the local currency in Ireland, Spain and elsewhere a significant exchange risk premium was payable prior to EMU as a cost of having an independent currency. This meant that the cost of capital for housing in such countries had been substantially greater than that for households in Germany and other countries where interest rates were broadly related to the DM. As this difference in the cost of capital had persisted over a long period it meant that prior to EMU, even with similar income levels and demographics, the optimal long-run stock of dwellings would have been lower than in Germany or the Netherlands. The advent of EMU, reducing the cost of capital permanently for households, will have raised the optimal long run stock of dwellings.

In addition to the interest rate premium payable by the household sector in Ireland and Spain prior to EMU, there was a further factor that is difficult to quantify, but which affected the availability of capital to fund housing investment. In the case of both Spain and Ireland we have seen in recent years that the surge in investment in housing has had its counterpart in a major increase in the balance of payments deficit. In turn, this deficit has been partially financed by the financial system borrowing abroad to onlend to the domestic household sector to fund their housing investment.





Prior to EMU the cost of funds on the Irish or Spanish interbank markets was affected by the supply and demand for funds in Ireland and Spain and the cost of foreign exchange. As discussed above there was a significant risk premium attached to lending in local currency relative to lending in DMs. Where traditional banks and building societies could fund their mortgage lending from local deposits the situation may have been similar in character to that in the 1970s and the 1980s. However, where the financial system was borrowing abroad to fund its lending, the exchange risk involved would have been passed through to borrowers. This would have involved both higher interest rates and the possibility of increased volatility.

Figure 3 shows the net foreign liabilities of the domestic financial system in Ireland, Spain and Poland relative to GDP. Over the last decade the housing booms in both Ireland and Spain relied on the availability of a ready supply of bank credit in the form of mortgages. If banks had behaved as in the 1970s and the 1980s they would have rationed mortgages while relying purely on domestic funds. To some extent this "rationing" might have shown up in an increase in the domestic interest rate unless monetary policy had proved accommodating. However, the liberalisation of the banking systems meant that banks were not constrained by their domestic deposit base, or even by the availability of funds on the local market.

As discussed above, in the 1980s and the early 1990s external borrowing by the domestic banking system would have involved a substantial foreign exchange risk premium. If banks had had to have recourse to this source of finance they would have been forced to pass on the cost of this premium to their customers. The resulting rise in interest rates would have slowed, if not choked off the housing booms in both

countries at an early stage. Such an outcome would also have prevented the housing booms reaching the level they did in both Spain and Ireland.

Instead, with the advent of EMU (and prior to the financial crisis), the domestic banking systems in both Ireland and Spain were able to raise substantial additional resources without any exchange risk and this provided the financing for a very large continuing investment in new building in both economies. In the case of Ireland this increase occurred very rapidly from 2003 onwards. With the collapse in house building in the Irish economy in 2008 (and in investment generally) the banks have begun the process of reducing their foreign borrowing.

The path of liberalisation of the Spanish financial system is rather different from that for Ireland. However, the 1990s also saw the development of a liberalised regime with a move away from credit rationing, as well as the move to EMU membership at the end of the decade.

The situation in Ireland and Spain in the period to 2008 was that banks had to raise very substantial funds abroad to finance the domestic housing boom. This was reflected in the dramatic rise since 2003 in the net foreign liabilities of the Irish domestic banking system shown in Figure 3. Because the funds are being raised in euros and the banks liabilities are also denominated in euros there is no exchange rate risk. In addition to raising additional funds on the Euro market, prior to the credit crisis it proved possible for banks to securitise mortgages and sell them on to the European market, reducing their exposure to the domestic market and raising additional funds to either finance more mortgage lending or investment in other assets.

The effect of EMU membership has thus been to reduce the financing costs for households engaging in a huge investment programme in domestic housing in Spain and Ireland. Prior to EMU such a boom would probably have fizzled out at a much earlier stage as banks faced an ever rising cost of funds, costs which they would have had to pass on to customers.

However, while EMU played a very important role in relaxing the constraints on the banking systems in Ireland and Spain the importance of the wider integration of financial markets should not be underestimated. The low cost of finance world-wide not only affected the Euro area but also the other members of the EU. As shown in Figure 3, Poland, while not a member of EMU saw a rather similar process, though on a more limited scale, where domestic banks borrowed abroad to fund domestic investment. However, unlike in Ireland, much of this funding may have been sourced within individual banks from their parents in Austria, Italy or Scandinavia. Nonetheless whatever the immediate source of the funds it did allow housing investment to expand in Poland.

A substantial part of this lending in Poland and Hungary took place in foreign currencies, especially Euros and Swiss Francs (Pawlowski, 2006). While this provided funding for housing to the household sector at a low nominal interest rate it also transferred the related exchange risk to those households. The fact that this source of funding was available at a low nominal interest rate in Poland (and other EU members not part of the Euro area) suggests that even without the advent of EMU, financial liberalisation could have still allowed significant housing booms in Ireland and Spain over the last decade.

The Real Cost of Capital

For firms selling their goods on the world market, whose output price is set on that market, the appropriate price deflator to use in calculating the real rate of interest is their output price. As a consequence differences in domestic inflation rates do not necessarily affect their cost of capital. Thus where a firm manufacturing cars in Spain, sells those cars at a price determined on the EU market in Germany, the Spanish rate of inflation in consumer prices does not directly affect its real cost of capital. Within EMU, with similar nominal interest rates across all EU members for similar firms the cost of capital for such firms is, as a result, independent of their location within the EMU and of the local rate of inflation in consumer prices.¹¹

However, for the household sector the appropriate price deflator is that for personal consumers' expenditure. As shown in Figure 4, in both Ireland and Spain real interest rates for households were very high in the early 1990s. However, with the approach of EMU real rates began to fall rapidly. While EMU was the major factor in the fall in interest rates, developments in global financial markets also played some role.

¹¹ Obviously higher rates of inflation in consumer prices may translate into growing labour cost differentials which would differentially affect firms in different countries in the EMU.



Figure 4: Real Interest Rates in Ireland and Spain

While EMU brought about a substantial reduction in the real cost of capital for households in many member countries, this reduction was particularly marked in the case of Spain and Ireland. With much higher rates of inflation in consumer prices in Ireland and Spain over the course of the first decade of EMU the real interest rate faced by the household sector in those countries was much lower than in Germany or France. In turn, this encouraged further investment in housing, adding to the pressure on the labour market and, ultimately, domestic consumer prices.

In addition to the reduction in the real interest rate for households, EMU made it much easier for the domestic financial system to fund a major investment surge by households. The fact that this change occurred as a result of a regime change (EMU) meant that it was perceived as a permanent reduction in the cost of capital raising the optimal long run stock of housing. When combined with the lower endowment of dwellings in Spain and Ireland it is not surprising that there was a strong impact on the housing sector.

5. The Role of Building in the Economy

The building and construction sector, and housing in particular, accounts for very different shares of the economy across EU member states. For those economies such as France and Germany, where the housing stock appears to be close to its long-term equilibrium, investment in housing accounts for around 5 to 6 per cent of GDP (Table 3). The average for the US is under 5 per cent. Over the last thirty five years, while there have been fluctuations in the share of housing investment in GDP, the difference in share between peak and trough has been between 3 and 4 percentage points of

GDP. This actually exaggerates the extent of the cyclical fluctuations for these countries as the peak occurred in the 1970s and the trough in the last decade. For the UK, housing investment accounts for an even smaller share of GDP. In the case of France and Germany, their recent investment in building and construction, expressed as a share of GDP, has been below the average for the last 35 years (Table 3). For the UK it has been slightly above the average, though well below the highest levels achieved in the past.

Table 3: Housing Investment as a share of GDP, 1970-2005, per cent								
	Housing			Other building				
Share of GDP	Average	Maximum	Minimum	2005	Average	Maximum	Minimum	2005
France	5.4	7.9	4.1	4.6	5.8	7.6	4.1	5.1
Germany	6.5	8.2	5.2	5.6	6.6	10.0	3.8	3.8
UK	3.6	4.5	2.8	3.9	5.2	7.0	4.3	5.0
Ireland	6.1	13.9	3.7	13.9	6.0	8.6	4.0	6.0
Spain	5.4	8.9	4.1	8.9	7.3	9.2	6.2	8.3
Poland, 1996-05	2.6	3.0	2.2	2.7	8.8	10.6	7.6	7.6

Figure 5: Housing Investment as a % of GDP¹²



The fact that housing accounts for a limited share of GDP does not prevent bubbles in house prices from occurring or from affecting aggregate economic activity through wealth effects but it does limit the direct effects of the housing cycle on aggregate demand. In the UK, in spite of the small share of economic activity accounted for by housing investment, it has suffered two cases of house price bubbles bursting in the last 20 years – one around 1990 and the second that is currently taking place.

In the case of Ireland and Spain the situation is very different, with the difference between the highest and lowest share of housing in GDP in Spain being just under 5

¹² In the case of Ireland it is as a percentage of GNP.

percentage points of GDP and a massive 10 per cent in the case of Ireland. In the case of both of these countries housing investment in 2006 accounted for a record share of GDP (Figure 5). This peak in activity is not just a cyclical event – the share of GDP accounted for by housing investment has been well above the previous levels over the past decade in both these countries reflecting both the demographic factors discussed above and also the changed circumstances of EMU. These two economies could not continue allocating such a high share of economic resources to this form of investment indefinitely.

The housing price and output booms in Spain and Ireland had a major impact on both these economies. To allocate such a high share of national resources to investment in housing, a major redeployment of resources was needed into the building and construction sector. High prices for housing and the enhanced profitability of the building sector resulted in a major expansion in activity, requiring a dramatic increase in the labour force employed in that sector. This bid up wage rates in the sector relative to the levels they would otherwise have maintained. In turn this attracted labour from other sector of the economy.¹³

While a significant part of the additional labour force in building and construction in both Ireland and Spain came from returning emigrants (in the case of Ireland), or from immigration in the case of both countries, there was, nonetheless, a major impact on the domestic labour market. At the peak approximately 20 per cent of employees in building and construction in Ireland were immigrants whereas the share in the late 1990s would have been close to zero. While this influx of labour moderated the impact on wage rates, it was not sufficient to offset the huge stimulus to the domestic labour markets in the two economies. The result has been that wage rates throughout the two economies rose much more rapidly than in the rest of EMU so that both economies lost competitiveness (Duffy et al., 2005).

The result of this demand shock has been that the tradable sector, in both Ireland and Spain, has been "crowded out" by the building and construction sector. Both countries have seen a rapid decline in market share for their exports and the result has

¹³ For example, in Ireland there has been a dramatic shift in the numbers training as engineers with the number of civil engineers growing dramatically and the number of electrical and electronic engineers falling. The reduced supply of electrical engineers, in turn, puts upward pressure on wages affecting sectors of the economy not involved in building and construction.

been a rapidly rising balance of payments deficit. This process is very similar to that where a major expansion in the public sector crowds out the (tradable) private sector. This process is described in Blanchard, 2007 where he illustrates the potential problems caused by this crowding out and concludes that government action may be required to moderate this process even in a perfect market (Blanchard, 2007).

This development would not necessarily have been very serious if the reallocation of resources to the building and construction sector were permanent. However, as argued above, the reallocation was essentially temporary due to a "catching up" in the stock of dwellings. If the catch up had been better managed by the respective governments so that it took place over a longer time scale then the reallocation of resources would have been more limited and the period for which the resources would have been more limited and the period for which the resources would have been more limited and construction sector would have been much longer. As it is, the collapse in housing output in Ireland and Spain is releasing major resources, especially labour, resulting in a very rapid rise in the unemployment rate. However, while many firms in the tradable sector were forced to shut down due to their loss of competitiveness over the last five years, it will be quite a number of years before the labour market adjusts to restore competitiveness. Even then it will be some time before domestic output of tradables adjusts to fill the gap ¹⁴.

As the adjustment continues, the balance of payments deficits in both countries can be expected to fall. However, in the case of Spain it is to be expected that in the long run there will be a continuing significant balance of payments deficit. Because so much of the demand for housing in Spain is for holiday homes and because many of these are bought by foreigners, the resulting capital inflow could sustain a substantial deficit over a prolonged period. While sales of holiday homes to foreigners are treated as a sale of an asset they are very close to being a tradable commodity. It is possible, if the market so desires, to build holiday homes very densely in large numbers for the foreseeable future – there is not a fixed supply of assets. This makes this market sustainable. The case is rather different for Estonia where much of medieval Tallinn has been sold to foreigners. As it takes a millennium to build a medieval dwelling such a flow of inward investment is not sustainable.

¹⁴ Bergin et al., 2009)

With approximately a third of Spanish dwellings being holiday homes they are financed by a substantial capital inflow. If all of these homes were owned and financed by foreigners (a considerable exaggeration) this would mean that the capital inflow related to such transactions would amount to one third of the investment in housing – around 3 per cent of GDP in 2006. Unlike investment in financial assets such investment in property is fairly illiquid which means that there is little danger of a reversal of the capital flow (though of course the inflow can dry up). In reality the capital inflow is probably less than this but it is still significant and must be taken into account in considering the size of the Spanish balance of payments deficit and its long-term sustainability.

Because of the size of the housing sector in Ireland and Spain the magnitude of the potential contraction is also very large. Whereas in the US and the UK a housing bust could see the fall in domestic demand directly caused by the contraction in housing amounting to two or at most three percentage points of GDP, in Ireland and Spain the potential contraction could amount to between 5 and 10 per cent of GDP.

The experience of Ireland and Spain holds out important lessons for other governments: there is a need to manage domestic housing markets using fiscal policy and financial regulation. For example, Poland currently has a low endowment of housing and is devoting a small share of national resources to new investment in the sector. This is not surprising. However, with rising incomes, financial liberalisation and membership of EMU Poland would be ripe for a housing boom. If not managed appropriately this could pre-empt resources needed for more productive uses as it has done in Spain and Ireland.

Chauvin and Le Bayon (2005) considered the housing markets in Spain, France, the UK and the US. Their conclusion was that of those four countries only in Spain was there a risk of a housing bubble. FitzGerald *et al.*, 2005 warned of the dangers of a housing bust in Ireland and modelled the possible impact of such an event. Cameron, et al (2006) reached a rather different conclusion for the UK suggesting that a significant risk of a bubble existing in that economy. Rae and van den Noord (2006), looking at the Irish economy suggested that house prices were significantly above their equilibrium level. In EUROFRAME-EFN, 2006, the exposure of the Irish and Spanish economies to shocks was considered. It was suggested in that report that in the case of both economies there was a risk of a housing bubble.

Throughout 2006 house prices continued to rise in Ireland, Spain and the UK and the building construction sector continued to increase its share of economic activity in the former two countries putting pressure on the tradable sector. As we know now the bubble has burst in all three economies (as well as in the US). The question is now what the impact of this will be on those economies and how long it will be before they return to "normal".

	Maximum Fall in Price
Denmark	-37
Finland	-50
France	-18
Germany	-15
Ireland ¹⁵	-27
Netherlands	-50
Sweden	-38
United Kingdom	-34
United States	-14

Table 4: International Experience of Real House Price Falls

Source: OECD, 2005 Economic Outlook, No. 78, November.

In all three economies the turning point came when potential buyers lost confidence in the prospect of future capital gains and stopped buying. The result was a fall in price and with the prospect of further falls in prices the user cost of owner occupation rose very rapidly. In turn, the building industry itself reacts with a significant lag so that output continued high in 2007. It is really only in 2008 in Ireland that the output of the sector began to show major adjustment.

What is interesting about the Spanish and Irish cases (and it is also true of the US) is that the bursting of the bubble was not caused by uncertainty in the financial markets but had a domestic trigger which changed agents expectations about future capital gains. Of course in the US it was the downturn in the housing market which caused (rather than was caused by) the financial dislocation that is so troubling the world economy. In the case of Ireland and Spain there is also very significant fall-out for the financial system from the burst property market bubble. In the case of the Irish financial system there was no exposure to toxic US assets. Instead the overexposure to risky domestic lending for property development has precipitated a major domestic financial crisis.

¹⁵ In Ireland the fall in real house prices was experienced between the third quarter of 1981 and the second quarter of 1987.

Table 4 shows the maximum falls in price that have occurred in the past in a sample of OECD economies when local housing bubbles burst. The largest falls experienced in the past were in Finland and the Netherlands where prices fell by 50 per cent. In quite a number of cases, including Ireland and the UK, the previous largest falls were of the order of 30 per cent. Past experience cannot tell us what will happen this time around in Ireland, Spain and the UK but it is a guide to the fact that the falls may be very significant (Kelly, 2007).

In the case of Ireland Van den Noord (2006) suggested an overvaluation of 15 per cent. Fitz Gerald et al., 2005, considered the potential for a fall in nominal house prices in Ireland of a third. However, while the equilibrium price might be 15 per cent (or even 30 per cent) below its peak level in 2006, prices may initially fall further than that. They will continue to fall until buyers are convinced that they can go no further down. As buyers return to the market they may then raise prices, which, by holding out the prospect of potential capital gains, in turn, reduces the user cost of housing thereby stimulating the market. Thus it is very difficult to forecast the trajectory of prices towards a new equilibrium as a bubble bursts.

The potential effects on Ireland of the housing bubble bursting were considered in detail in Fitz Gerald, *et al.*, 2005. Since that analysis was carried out the housing sector continued to expand for one further year, buoyed up by inappropriately stimulatory government policy. Thus the potential fall in output could be larger than that work envisaged. However, it does give a reasonable quantification of the potential impact of the bursting of the housing bubble on the economy.

However, for both Spain and Ireland the bursting of the property market bubble has coincided with the world recession and this has greatly aggravated the effects on the two economies. In the case of Ireland the cumulative fall in GDP over the period 2008-10 is expected to be around 12 per cent (Barrett *et al.*, 2009) while the fall in Spain could be of the order of 3 per cent (Holland *et al.*, 2009). Rising unemployment will result in a major slowdown in the rate of increase in labour costs, especially in Ireland. (Current forecasts using a model of the labour market suggest that nominal wage rates could fall by 7 per cent between 2009 and 2011, Bergin *et al.*, 2009.) If such an adjustment in labour costs does take place it will eventually result in an expansion in the tradable sector of the economy as exports rise consistent with a more competitive environment. However, there is likely to be a major permanent reduction

in Irish output as a result of the combined impact of the bursting of the property market bubble and the world recession.

In terms of the macro-economic dislocation consequent on the bursting of the housing bubble, a very important factor will be how long the adjustment to a new equilibrium takes. Typically, in the OECD area it takes between one and two years for sellers to accept a fall in real house prices of 10 percentage points (Kelly, 2007). This would suggest that if nominal house prices were to fall by a third relative to their peak in Ireland it could take between three and four years.¹⁶

The longer drawn out the adjustment process the greater the economic cost. While sellers refuse to contemplate a major fall in price and potential buyers hold out of the market there is no incentive for anyone to build more dwellings. In addition, all the businesses which are closely allied to the housing market (mortgage lenders, lawyers, estate agents, suppliers of furniture etc.) experience a dramatic downturn in activity. If sellers could be persuaded to adjust their expectations much more rapidly, so that prices find their long-term equilibrium level, the housing sector would see an earlier return to "normal". In the case of the UK today the adjustment of expectations seems unusually rapid, which may hasten the end of the crisis. In Ireland house prices seem to be falling at the rate of around 10 per cent a year.

Another measure of the extent of the adjustment needed is the excess stock of unoccupied dwellings. It is difficult to estimate such a number but it is clear that there is a build up in stocks of finished dwellings in both Ireland and Spain. Until these dwellings are filled through renting or sale to new households there will continue to be downward pressure on the market. In both countries the demographics mean that there will be a continuing rise in household numbers in spite of the slowdown. Thus, even with inertia, the excess stock will be gradually eaten away. However, it is likely that the dynamics of the market will produce a somewhat faster resolution.

Pressures from the banking system to see an immediate return on the assets has resulted in a significant part of the excess stock being put on the rental market. There is strong evidence that this is happening with a consequential major reduction in rents. As outlined earlier, the rental market is very thin in Ireland and Spain and such an outcome would have a major initial downward impact on rents. Already renting is

¹⁶ This takes account of the underlying rate of inflation in the economy.

more attractive than purchase at current house prices. With rents falling and no prospect in the next year of a rise in house prices there is every incentive for households to rent rather than buy. However, the fall in the cost of housing will result in a rise in demand. In particular, the much lower headship rates in Ireland and Spain than might be expected means that there is considerable scope for headship rates to rise. Bergin *et al.*, 2009, suggests that this is already happening in Ireland in spite of the rising unemployment rate. Once the non-housing economy shows signs of recovering there could be a rapid take-up of "surplus" rented dwellings and a move by Irish and Spanish households to a more "European" rate of household formation.

6. Managing the housing market under EMU

Since the mid-1990s there has been considerable debate about how fiscal policy should be used within EMU (Butti *et al.*, 2003). A lot of attention has been given to the Stability and Growth Pact (SGP) and the need to prevent government deficits exceeding 3 per cent of GDP. However, much less attention has been given to how fiscal policy should be used to manage individual regional economies. In particular, the potential problems that may arise for regional economies in EMU from local asset market bubbles, primarily in property, have not received much attention. We know now that the costs of failing to control such phenomena are very high and can be magnified by the damage done to the regional financial system.¹⁷

Generally, where substantial bubbles do occur they are reflected in a significant balance of payments deficit. In the case of Spain the rising balance of payments deficit goes back to the early years of EMU whereas for Ireland it only manifested itself in the last five years. In both cases the balance of payments deficit was a symptom of the domestic crowding out of the tradable sector due to the dramatic increase in building activity beyond its sustainable level. The consequence was a rapid rise in domestic costs, pricing the tradable sector out of its world markets.

¹⁷ The potential damage done to regional economies from bursting property market bubbles is also significant for regions in the US. However, in the US the attention has focused more on using financial regulation rather than fiscal policy to manage property markets.



Looking at Ireland and Spain in 2001, Blanchard, 2001, suggested that significant balance of payments deficit then apparent in Spain suggested a need to tighten fiscal policy whereas for Ireland, running a balance of payments surplus, a real appreciation of the exchange rate was appropriate. However, on this basis, as Ireland moved into balance of payments deficit in 2003, a similar prescription would have been appropriate. While both countries complied with the rules of the Stability and Growth Pact running small surpluses this was not adequate to protect their own economies from the risk of overheating. Instead what was required was a policy similar to that adopted in Scandinavia where fiscal policy ran a substantial surplus preventing serious overheating. Figure 6 shows the general government balance in Ireland, Spain and Finland in recent years. As a result of a more prudent fiscal policy Finland now finds itself better equipped to face the current economic crisis. Thus within EMU it may be prudent for countries to run substantial surpluses for a number of years where domestic inflationary pressures are posing problems.

While EMU may not make an EMU-wide housing bubble any more likely, the restriction on the use of the interest rate to manage the domestic housing market increases risks from housing shocks for individual economies. As a result, monetary policy, which was potentially an effective tool for managing national housing market is no longer available for that purpose. There are, therefore, concerns, that the authorities in individual countries such as Ireland and Spain are handicapped in managing potential risks to their economic stability arising from housing market

developments. However, there are other instruments available to governments in EMU to achieve the same effect. The problem is just that they have not been used.

In preparation for possible membership of EMU the management of the housing sector was considered by the British Treasury in a document published in 2003 (Treasury, 2003). They considered that the loss of the interest rate instrument would obviously restrict the scope for economic policy to manage this important sector of the economy. They suggested that policy makers would have to make more active use of fiscal instruments, specifically taxation, to manage risks to the wider economy arising from possible housing bubbles. In particular, they suggested that stamp duty (a transactions tax) could be used to temporarily change the cost of housing investment for households. Fitz Gerald, 2001, suggested reducing income tax allowances for mortgage interest payments or even a tax on mortgage interest payments.

In each case the effect of the tax instrument would be to raise the cost of borrowing or financing housing investment, tending to reduce demand pressures and, hence, prices below the level they would otherwise reach. As outlined in the model in Section 3, by raising the cost of capital such a tax would exert significant downward pressure on house prices. Through its effect on the income of potential borrowers it would also tend to reduce demand pressures. Of the alternatives the taxation of mortgage interest payments may be the best instrument to use. In increasing the current outgoings of households it would reduce their ability to take on debt. As a result, when the danger of a bubble is over and the taxes are abolished or reduced, households would end up with a lower stock of debt for the same stock of housing. However, adopting such targeted fiscal policy instruments may be difficult to explain to a wider public.

Within the Euro area the risks to the financial sector from any potential, housing bubble depend on the extent of the regional banking system's exposure to the regional economy and its exposure within that economy to the housing (property) sector.¹⁸ Because of the fact that housing bubbles are essentially regional in nature, the more geographically diversified the financial system is the more robust it will be in the face

¹⁸ Here we are only considering the direct exposure of the regional banking system to the local market. As we have seen the exposure of the European banking system to the US sub-prime market was very substantial. This risk transfer occurred through the use of complicated financial instruments. The regulation of such instruments is a separate issue not considered here.

of idiosyncratic shocks. However, it is important for the regulatory authorities when stress testing their financial systems to take account of both the regional diversity of a bank's loan portfolio and also to take account of the possible wider economic consequences of a housing bubble. In particular, when housing bubbles burst they tend to be associated with much wider macro-economic disruption, which is likely to affect employment and output levels.

A serious shock to the agricultural, energy and regional property sectors in the Federal Reserve of Kansas City District in the US in the mid-1980s caused many local banks to fail.¹⁹ As a result of this experience the local Federal Reserve Board advocated greatly increased securitisation of mortgage debt in its *Economic Review* in subsequent years. This was designed to reduce the vulnerability of the local financial system, which was very exposed to the regional economy (due to lack of regional diversification due to the operation of the Glass-Stiegel act). However, we have seen the dangers that securitisation can bring: the securitisation of mortgage assets in the US spread the risks of a housing bust far and wide. This highlighted the fact that securitisation is only as good as the original mortgage assets and we have seen that in the case of the US these assets were not very good! However, if properly regulated such instruments may still prove important in reducing risks where banking systems are regional in nature.

The financial regulatory authorities in individual regional economies can play a vital role in ensuring the financial stability of their financial systems. If more attention had been paid to the threats to financial stability in Ireland the problems today would be much less acute than they actually are. However, in a financially integrated world, where banks operating in an individual economy are regulated by different national authorities, it may not be feasible to manage the domestic housing market merely by better financial regulation. In the case of Ireland, even if the domestic regulator had acted sensibly, limiting the exposure of the domestic financial system to the domestic housing market, this would not have prevented foreign owned banks from providing excessive credit. The only comprehensive instrument available to the government was fiscal policy.

¹⁹ 63 banks failed in the region in 1985 and 69 in 1986, Keeton and Hecht, 1987.

Outside the Euro area, in particular in Poland and Hungary (see EFN, 2006), the potential risks to the economy from a housing boom are being enhanced through the move to borrowing in euros or Swiss francs. Such borrowing by households in foreign currencies was previously a factor in the financial crisis in Scandinavia in the early 1990s (Jonung, Kiander and Vartia, 2008). This pattern of behaviour makes these economies vulnerable to exchange rate shocks and it could also see the costs to the economy of a normal adjustment process after a macro-economic shock, involving changes in exchange rates, being greatly magnified through the direct effects on the net asset position of households.

The experience of the past, both in the US and in Europe, indicates that there is a little danger of a Euro-wide housing bubble posing problems for policy makers in the near future. However, regional booms and busts can and are happening and past experience indicates that they have serious implications for the regional economies so affected. In addition, problems in the housing market can affect the regional financial sector where problems of bad debts occur. This emphasises the need for the authorities in individual EU member states to use fiscal policy actively to prevent bubbles occurring. They also need to improve their regulation of their domestic financial systems to discourage banks from taking excessive risk which endangers the economy as a whole as well, of course, as the individual banks' well-being.

7. Conclusions

Prior to EMU member states had the opportunity to manage the domestic housing market through monetary policy. However, under EMU monetary policy is targeted at the Euro area inflation rate and, because of the idiosyncratic nature of the housing sector across the Euro zone, the stance of monetary policy will only help control housing bubbles by accident.

Under these circumstances the best instrument available to governments to manage regional housing markets is fiscal policy. Through suitably targeted tax instruments the authorities can change the cost of housing services faced by households, influencing their investment behaviour. Using this instrument effectively may prove politically difficult, partly because of its novelty. However, as housing bubbles are as likely to happen in the future as in the past, it will be important for governments to develop an understanding of the potential dangers of shocks in the housing sector to the wider economy and of the implications of EMU for how the housing market must be managed.

In addition, the advent of monetary union has not absolved governments from the duty to manage domestic inflationary pressures using fiscal policy. Instead of focusing on the SGP target of "not running a deficit" it may instead be appropriate for some countries to run substantial surpluses for a period of years.

A second important implication for policy is the need for the regulatory authorities to take appropriate measures to safeguard their domestic financial sectors. In stress testing the financial system they need to take account of the likelihood that macroeconomic shocks may affect all economies within the EMU. Such shocks, whether or not they are combined, with a rise in real interest rates, may seriously affect the housing sector. While shocks to the housing sector are most unlikely to be generalised across the Euro area, they may affect a number of regional economies simultaneously, such as in Scandinavia in the late 1980s and early 1990s. Under these circumstances the robustness of individual bank's loan portfolios needs to be tested by shocks which might simultaneously affect a number of the markets in which the individual banks trade.

References

- Baker, T. J. Fitz Gerald and P. Honohan, 1996, *Economic Implications for Ireland of EMU*, Policy Research Series No. 28, The Economic and Social Research Institute, Dublin.
- Barrett, A., J. Goggin and I. Kearney, 2009. Quarterly Economic Commentary, Spring, Dublin: The Economic and Social Research Institute.
- Bergin A., T. Conefrey, J. FitzGerald and I. Kearney, 2009, *Recovery Scenarios for Ireland*, Dublin: The Economic and Social Research Institute, Research Series No. 7.
- Blanchard, O., 2001, "Country adjustments within the euro area: lessons after two years". In *Defining a Macroeconomic Framework for the Euro Area*, London: CEPR.
- Blanchard, O., 2007, "Current account deficits in rich countries", NBER Working paper No. 12925.
- Butti, M., 2003, *Monetary and Fiscal Policies in EMU*, Cambridge: Cambridge University Press.
- Calmfors, L. ed., 1997, *EMU: A Swedish Perspective*, Boston: Kluwer Academic Publishers.
- Cameron, G., J. Muellbauer and A. Murphy, 2006, "Was there a British House Price Bubble? Evidence from a Regional Panel", London CEPR 5619
- Chauvin, V. and S. Le Bayon., 2005, "Logement: Sommets Atteints?", Lettre de l'OFCE, No. 257

- Duffy, 2002, "A Descriptive Analysis of the Irish Housing Market", Dublin: ESRI *Quarterly Economic Commentary*, July.
- Duffy, D., Fitz Gerald, J. and I. Kearney, 2005, "Rising House Prices in an Open Labour Market", *Economic and Social Review*, Vol 36 No 3, Winter
- ECB, 2006, "Assessing house price developments in the Euro area", ECB Monthly Bulletin, February.
- EUROFRAME European Forecasting Network, 2006, "Economic Forecasts of the Euro Area: Forecasts and Policy Analysis", Dublin: The ESRI, Spring.
- Faust, J., Rogers, J. H., & Wright, J. H. 2001. An Empirical Comparison of Bundesbank and ECB Monetary Policy Rules. *Federal Reserve System*, International Finance Discussion Papers, No 705.
- Fitz Gerald, J., 2001, "Fiscal Policy in a Monetary Union: the Case of Ireland", *Quarterly Economic Commentary*, March.
- Fitz Gerald J., A. Bergin, I. Kearney, A. Barrett, D. Duffy, S. Garrett, and Y. McCarthy, 2005, *Medium-Term Review: 2005-2012*, Dublin: The Economic and Social Research Institute.
- Fitz Gerald J., *et al.*, 2008, *Medium-Term Review: 2008-2015*, Dublin: The Economic and Social Research Institute.
- Garcia, J. A., and J. E Rodriguez Hernandez, 2008, "Housing Demand in Spain according to dwelling type: Microeconometric evidence", *Regional Science and Urban Economics*, vol. 38, pp. 363-377.
- Girouard, N., M. Kennedy, P. van den Noord and C. Andre, 2006, "Recent House Price Developments: The Role of Fundamentals", OECD Economics Department Working Paper No. 475.
- Gonzalez, L. and F Ortega, 2009, "Immigration and Housing booms: Evidence from Spain", IZA Discussion Paper No. 4333.
- Himmelberg, C., C. Mayer and T. Sinai, 2005, "Assessing High House Prices: Bubbles, Fundamentals, and Misperceptions", *Journal of Economic Perspectives*, Vol. 19, No.4.
- Holland D., R. Barrell, T. Fic, I. Hurst, I Liadze., A. Orzagani and V. Pillonca, "The World Economy", *National Institute Economic Review*, No. 208, April.
- Honohan, P. and D. Klingebiel, 2003, "The Fiscal Cost Implications of an Accommodating Approach to Banking Crises" *Journal of Banking and Finance*. 2003.
- Jonung, L., J. Kiander and P. Vartia, 2008. "The great crisis in Finland and Sweden The macrodynamics of boom, bust and recovery 1985-2000. Report for the study Crises, macroeconomic performance and economic policies in Finland and Sweden in the 1990s: A comparative approach", Working paper, Brussels: DG ECFIN.
- Keeton, W., and K. Hecht, 1987, "Banking Performance in tenth District States", *Federal Reserve Bank of Kansas City Economic Review*, July-August, Volume 72, Number 7
- Kelly, M., 2007, "On the Likely Extent of Falls in Irish House Prices", Dublin: ESRI *Quarterly Economic Commentary*, Summer.
- Kenny, G., 1999. "Modelling the Demand and Supply Sides of the Housing Market: Evidence from Ireland", *Economic Modelling*, 16, pp. 389-409.
- McQuinn, K. 2004. "A Model of the Irish Housing Sector", *Central Bank and Financial Services Authority of Ireland*, Research Paper 1/RT/04.

- Muellbauer, J. and A. Murphy, 2008, "Economic Influences on International Migration for the UK", Oxford University mimeo.
- Murphy, A. 1998. "Econometric Modelling of the Irish Housing Market", in P. Bacon, F. MacCabe, & A. Murphy, eds. (eds.), An Economic Assessment of Recent House Price Developments, Dublin: Government of Ireland Publications, Stationery Office.
- Murphy, 2006, "Modelling Irish House Prices: A Review and Some New Results", Nuffield College Oxford mimeo.
- OECD, 2005, Economic Outlook, Paris: OECD.
- Pawlowski, 2006, "Financial System Stability in the New EU Member States Ahead of the Euro Adoption", Appendix to Euroframe, *Economic Assessment of the Euro Area*, Spring Report 2006.
- Rae, D. and P. van den NOORD, 2006, "Ireland's Housing Boom: What has driven it and have prices overshot?, Paris: OECD Economics Department Working Paper No. 492.
- Roche, M. 2001. "The Rise in House Prices in Dublin: Fad or Just Fundamentals?", *Economic Modelling*, 18, pp. 281-295.
- Roche, M. 2003. "Will there be a Crash in Irish House Prices?", *Quarterly Economic Commentary*, Winter, pp. 57-72.

Treasury, HM, 2003, Fiscal Stabilisation and EMU, London: HM Treasury.

van den NOORD, P., 2006 "Are House Prices Nearing a Peak? A Probit Analysis for 17 Countries", Paris: OECD Economics Department Working Paper No. 88.

Appendix 1

(1). Inverted demand equation for new house prices: Ireland

$$Log\left(\frac{PHNEW}{PC}\right)_{t} = 8.41 + 0.99 \log\left(\frac{Y}{POP}\right) - 3.79 \log\left(1 + \frac{HSTOCK}{POP}\right)$$
$$+ 14.12\left(\frac{POP \ 2534}{POP}\right) - 0.97(rr) + 0.11(D03)$$

(2). Inverted demand equation for new house prices: Spain

$$Log\left(\frac{PHNEW}{PC}\right)_{t} = -6.71 + 1.13 \log\left(\frac{Y}{POP}\right) - 8.22 \log\left(\frac{1 + HSTOCK}{POP}\right)$$

+ 25.61
$$\left(\frac{POP\ 2534}{POP}\right)$$
 - 0.85 log (rr) + 0.11 $(D\ 87)$ + 0.27 $D(03)$

Dependent variable: Log	Ireland (1974 – 2006)		Spain (1978 - 2005)		
Variables in model:	Description:	Coefficients:	t-statistic	Coefficients:	t-statistic
С	Intercept	8.41	56.78	-6.71	-2.51
LOG(Y)	Log (Income per capita)	0.99	6.56	1.13	2.54
LOG(HSTOCK)	Log (housing stock per capita)	-3.79	-2.67	-8.22	-3.73
POP2534	Household formation ratio	14.12	6.96	25.61	2.93
LOG(rr)	User cost lagged one year	-0.97	-4.33	-0.85	-5.23
D87	Dummy from 1987 onwards			0.11	1.89
D03	Dummy from 2003 onwards	0.11	3.16	0.27	5.08
Number of observation	IS	33		28	
R ²		0.99		0.99	
Adjusted R ²			0.99		0.98
		Test statistic:		Test statistic:	
S.E. of regression		0.04		0.05	
Sum squared resid		0.05		0.06	
Log likelihood		60.41		47.43	
F-statistic		503.08		239.78	
S.D. dependent var		0.38		0.38	
Akaike info criterion		-3.30		-2.89	
Durbin-Watson stat		1.26		1.53	

Year	Number	Title/ Author(s) ESRI Authors/ Co-authors <i>Italicised</i>
2009		
	314	Job Mismatches and Labour Market Outcomes Kostas Mavromaras, <i>Seamus McGuinness,</i> Nigel O'Leary, Peter Sloane, and Yin King Fok
	313	Immigrants and Employer-provided Training <i>Alan Barrett, Séamus McGuinness, Martin O'Brien</i> and <i>Philip O'Connell</i>
	312	Did the Celtic Tiger Decrease Socio-Economic Differentials in Perinatal Mortality in Ireland? <i>Richard Layte</i> and <i>Barbara Clyne</i>
	311	Exploring International Differences in Rates of Return to Education: Evidence from EU SILC Maria A. Davia, <i>Seamus McGuinness</i> and <i>Philip, J.</i> <i>O'Connell</i>
	310	Car Ownership and Mode of Transport to Work in Ireland <i>Nicola Commins</i> and <i>Anne Nolan</i>
	309	Recent Trends in the Caesarean Section Rate in Ireland 1999-2006 <i>Aoife Brick</i> and <i>Richard Layte</i>
	308	Price Inflation and Income Distribution Anne Jennings, Seán Lyons and Richard S.J. Tol
	307	Overskilling Dynamics and Education Pathways Kostas Mavromaras, <i>Seamus McGuinness</i> , Yin King Fok
	306	What Determines the Attractiveness of the European Union to the Location of R&D Multinational Firms? <i>Iulia Siedschlag, Donal Smith, Camelia Turcu, Xiaoheng Zhang</i>
	305	Do Foreign Mergers and Acquisitions Boost Firm Productivity? <i>Marc Schiffbauer, Iulia Siedschlag, Frances Ruane</i>
	304	Inclusion or Diversion in Higher Education in the Republic of Ireland? <i>Delma Byrne</i>
	303	Welfare Regime and Social Class Variation in Poverty and Economic Vulnerability in Europe: An Analysis of EU- SILC

	Christopher T. Whelan and Bertrand Maître
302	Understanding the Socio-Economic Distribution and Consequences of Patterns of Multiple Deprivation: An Application of Self-Organising Maps Christopher T. Whelan, Mario Lucchini, Maurizio Pisati and <i>Bertrand Maître</i>
301	Estimating the Impact of Metro North Edgar Morgenroth
300	Explaining Structural Change in Cardiovascular Mortality in Ireland 1995-2005: A Time Series Analysis <i>Richard Layte, Sinead O'Hara</i> and Kathleen Bennett
299	EU Climate Change Policy 2013-2020: Using the Clean Development Mechanism More Effectively <i>Paul K Gorecki, Seán Lyons</i> and <i>Richard S.J. Tol</i>
298	Irish Public Capital Spending in a Recession Edgar Morgenroth
297	Exporting and Ownership Contributions to Irish Manufacturing Productivity Growth Anne Marie Gleeson, <i>Frances Ruane</i>
296	Eligibility for Free Primary Care and Avoidable Hospitalisations in Ireland <i>Anne Nolan</i>
295	Managing Household Waste in Ireland: Behavioural Parameters and Policy Options <i>John Curtis, Seán Lyons</i> and <i>Abigail O'Callaghan-Platt</i>
294	Labour Market Mismatch Among UK Graduates; An Analysis Using REFLEX Data <i>Seamus McGuinness</i> and <i>Peter J. Sloane</i>
293	Towards Regional Environmental Accounts for Ireland <i>Richard S.J. Tol , Nicola Commins, Niamh Crilly, Sean</i> <i>Lyons</i> and <i>Edgar Morgenroth</i>
292	EU Climate Change Policy 2013-2020: Thoughts on Property Rights and Market Choices <i>Paul K. Gorecki, Sean Lyons</i> and <i>Richard S.J. Tol</i>
291	Measuring House Price Change David Duffy
290	Intra-and Extra-Union Flexibility in Meeting the European Union's Emission Reduction Targets

Richard S.J. Tol

289	The Determinants and Effects of Training at Work: Bringing the Workplace Back In <i>Philip J. O'Connell</i> and <i>Delma Byrne</i>
288	Climate Feedbacks on the Terrestrial Biosphere and the Economics of Climate Policy: An Application of <i>FUND Richard S.J. Tol</i>
287	The Behaviour of the Irish Economy: Insights from the HERMES macro-economic model <i>Adele Bergin, Thomas Conefrey, John FitzGerald</i> and <i>Ide Kearney</i>
286	Mapping Patterns of Multiple Deprivation Using Self-Organising Maps: An Application to EU-SILC Data for Ireland Maurizio Pisati, <i>Christopher T. Whelan</i> , Mario Lucchini and <i>Bertrand Maître</i>
285	The Feasibility of Low Concentration Targets: An Application of FUND <i>Richard S.J. Tol</i>
284	Policy Options to Reduce Ireland's GHG Emissions Instrument choice: the pros and cons of alternative policy instruments Thomas Legge and <i>Sue Scott</i>
283	Accounting for Taste: An Examination of Socioeconomic Gradients in Attendance at Arts Events <i>Pete Lunn</i> and <i>Elish Kelly</i>
282	The Economic Impact of Ocean Acidification on Coral Reefs Luke M. Brander, Katrin Rehdanz, <i>Richard S.J. Tol</i> , and Pieter J.H. van Beukering
281	Assessing the impact of biodiversity on tourism flows: A model for tourist behaviour and its policy implications Giulia Macagno, Maria Loureiro, Paulo A.L.D. Nunes and <i>Richard S.J. Tol</i>
280	Advertising to boost energy efficiency: the Power of One campaign and natural gas consumption <i>Seán Diffney, Seán Lyons</i> and <i>Laura Malaguzzi Valeri</i>
279	International Transmission of Business Cycles Between Ireland and its Trading Partners <i>Jean Goggin</i> and <i>Iulia Siedschlag</i>

	278	Optimal Global Dynamic Carbon Taxation David Anthoff
	277	Energy Use and Appliance Ownership in Ireland <i>Eimear Leahy</i> and <i>Seán Lyons</i>
	276	Discounting for Climate Change <i>David Anthoff, Richard S.J. Tol</i> and Gary W. Yohe
	275	Projecting the Future Numbers of Migrant Workers in the Health and Social Care Sectors in Ireland <i>Alan Barrett</i> and Anna Rust
	274	Economic Costs of Extratropical Storms under Climate Change: An application of FUND Daiju Narita, <i>Richard S.J. Tol, David Anthoff</i>
	273	The Macro-Economic Impact of Changing the Rate of Corporation Tax <i>Thomas Conefrey</i> and <i>John D. Fitz Gerald</i>
	272	The Games We Used to Play An Application of Survival Analysis to the Sporting Life- course Pete Lunn
2008		
	271	Exploring the Economic Geography of Ireland Edgar Morgenroth
	270	Benchmarking, Social Partnership and Higher Remuneration: Wage Settling Institutions and the Public-Private Sector Wage Gap in Ireland <i>Elish Kelly, Seamus McGuinness, Philip O'Connell</i>
	269	A Dynamic Analysis of Household Car Ownership in Ireland <i>Anne Nolan</i>
	268	The Determinants of Mode of Transport to Work in the Greater Dublin Area <i>Nicola Commins</i> and <i>Anne Nolan</i>
	267	Resonances from <i>Economic Development</i> for Current Economic Policymaking <i>Frances Ruane</i>
	266	The Impact of Wage Bargaining Regime on Firm-Level Competitiveness and Wage Inequality: The Case of Ireland <i>Seamus McGuinness, Elish Kelly</i> and <i>Philip O'Connell</i>

265	Poverty in Ireland in Comparative European Perspective <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
264	A Hedonic Analysis of the Value of Rail Transport in the Greater Dublin Area <i>Karen Mayor, Seán Lyons, David Duffy</i> and <i>Richard S.J.</i> <i>Tol</i>
263	Comparing Poverty Indicators in an Enlarged EU <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
262	Fuel Poverty in Ireland: Extent, Affected Groups and Policy Issues <i>Sue Scott, Seán Lyons, Claire Keane,</i> Donal McCarthy and <i>Richard S.J. Tol</i>
261	The Misperception of Inflation by Irish Consumers <i>David Duffy</i> and <i>Pete Lunn</i>
260	The Direct Impact of Climate Change on Regional Labour Productivity Tord Kjellstrom, R Sari Kovats, Simon J. Lloyd, Tom Holt, <i>Richard S.J. Tol</i>
259	Damage Costs of Climate Change through Intensification of Tropical Cyclone Activities: An Application of FUND Daiju Narita, <i>Richard S. J. Tol</i> and <i>David Anthoff</i>
258	Are Over-educated People Insiders or Outsiders? A Case of Job Search Methods and Over-education in UK Aleksander Kucel, <i>Delma Byrne</i>
257	Metrics for Aggregating the Climate Effect of Different Emissions: A Unifying Framework <i>Richard S.J. Tol,</i> Terje K. Berntsen, Brian C. O'Neill, Jan S. Fuglestvedt, Keith P. Shine, Yves Balkanski and Laszlo Makra
256	Intra-Union Flexibility of Non-ETS Emission Reduction Obligations in the European Union <i>Richard S.J. Tol</i>
255	The Economic Impact of Climate Change Richard S.J. Tol
254	Measuring International Inequity Aversion <i>Richard S.J. Tol</i>
253	Using a Census to Assess the Reliability of a National Household Survey for Migration Research: The Case of

Ireland Alan Barrett and Elish Kelly

252	Risk Aversion, Time Preference, and the Social Cost of Carbon <i>David Anthoff, Richard S.J. Tol</i> and Gary W. Yohe
251	The Impact of a Carbon Tax on Economic Growth and Carbon Dioxide Emissions in Ireland <i>Thomas Conefrey, John D. Fitz Gerald, Laura Malaguzzi</i> <i>Valeri</i> and <i>Richard S.J. Tol</i>
250	The Distributional Implications of a Carbon Tax in Ireland <i>Tim Callan, Sean Lyons, Susan Scott, Richard S.J. Tol</i> and Stefano Verde
249	Measuring Material Deprivation in the Enlarged EU <i>Christopher T. Whelan, Brian Nolan</i> and <i>Bertrand Maître</i>
248	Marginal Abatement Costs on Carbon-Dioxide Emissions: A Meta-Analysis Onno Kuik, Luke Brander and <i>Richard S.J. Tol</i>
247	Incorporating GHG Emission Costs in the Economic Appraisal of Projects Supported by State Development Agencies <i>Richard S.J. Tol</i> and <i>Seán Lyons</i>
246	A Carton Tax for Ireland <i>Richard S.J. Tol, Tim Callan, Thomas Conefrey, John D.</i> <i>Fitz Gerald, Seán Lyons, Laura Malaguzzi Valeri</i> and Sugan Seatt
245	Non-cash Benefits and the Distribution of Economic Welfare <i>Tim Callan</i> and <i>Claire Keane</i>
244	Scenarios of Carbon Dioxide Emissions from Aviation <i>Karen Mayor</i> and <i>Richard S.J. Tol</i>
243	The Effect of the Euro on Export Patterns: Empirical Evidence from Industry Data <i>Gavin Murphy</i> and <i>Iulia Siedschlag</i>
242	The Economic Returns to Field of Study and Competencies Among Higher Education Graduates in Ireland <i>Elish Kelly, Philip O'Connell</i> and <i>Emer Smyth</i>
241	European Climate Policy and Aviation Emissions Karen Mayor and Richard S.J. Tol

240	Aviation and the Environment in the Context of the EU- US Open Skies Agreement <i>Karen Mayor</i> and <i>Richard S.J. Tol</i>
239	Yuppie Kvetch? Work-life Conflict and Social Class in Western Europe <i>Frances McGinnity</i> and <i>Emma Calvert</i>
238	Immigrants and Welfare Programmes: Exploring the Interactions between Immigrant Characteristics, Immigrant Welfare Dependence and Welfare Policy <i>Alan Barrett</i> and Yvonne McCarthy
237	How Local is Hospital Treatment? An Exploratory Analysis of Public/Private Variation in Location of Treatment in Irish Acute Public Hospitals Jacqueline O'Reilly and Miriam M. Wiley
236	The Immigrant Earnings Disadvantage Across the Earnings and Skills Distributions: The Case of Immigrants from the EU's New Member States in Ireland <i>Alan Barrett, Seamus McGuinness</i> and <i>Martin O'Brien</i>
235	Europeanisation of Inequality and European Reference Groups <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
234	Managing Capital Flows: Experiences from Central and Eastern Europe Jürgen von Hagen and <i>Iulia Siedschlag</i>
233	ICT Diffusion, Innovation Systems, Globalisation and Regional Economic Dynamics: Theory and Empirical Evidence Charlie Karlsson, Gunther Maier, Michaela Trippl, <i>Iulia</i> <i>Siedschlag,</i> Robert Owen and <i>Gavin Murphy</i>
232	Welfare and Competition Effects of Electricity Interconnection between Great Britain and Ireland <i>Laura Malaguzzi Valeri</i>
231	Is FDI into China Crowding Out the FDI into the European Union? Laura Resmini and <i>Iulia Siedschlag</i>
230	Estimating the Economic Cost of Disability in Ireland John Cullinan, Brenda Gannon and <i>Seán Lyons</i>
229	Controlling the Cost of Controlling the Climate: The Irish Government's Climate Change Strategy Colm McCarthy, <i>Sue Scott</i>

	228	The Impact of Climate Change on the Balanced-Growth- Equivalent: An Application of <i>FUND</i> <i>David Anthoff</i> , <i>Richard S.J. Tol</i>
	227	Changing Returns to Education During a Boom? The Case of Ireland <i>Seamus McGuinness</i> , <i>Frances McGinnity, Philip O'Connell</i>
	226	'New' and 'Old' Social Risks: Life Cycle and Social Class Perspectives on Social Exclusion in Ireland <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
	225	The Climate Preferences of Irish Tourists by Purpose of Travel <i>Seán Lyons, Karen Mayor</i> and <i>Richard S.J. Tol</i>
	224	A Hirsch Measure for the Quality of Research Supervision, and an Illustration with Trade Economists <i>Frances P. Ruane</i> and <i>Richard S.J. Tol</i>
	223	Environmental Accounts for the Republic of Ireland: 1990-2005 <i>Seán Lyons, Karen Mayor</i> and <i>Richard S.J. Tol</i>
2007	222	Assessing Vulnerability of Selected Sectors under Environmental Tax Reform: The issue of pricing power <i>J. Fitz Gerald</i> , M. Keeney and <i>S. Scott</i>
	221	Climate Policy Versus Development Aid Richard S.J. Tol
	220	Exports and Productivity – Comparable Evidence for 14 Countries <i>The International Study Group on Exports and</i> <i>Productivity</i>
	219	Energy-Using Appliances and Energy-Saving Features: Determinants of Ownership in Ireland Joe O'Doherty, <i>Seán Lyons</i> and <i>Richard S.J. Tol</i>
	218	The Public/Private Mix in Irish Acute Public Hospitals: Trends and Implications Jacqueline O'Reilly and Miriam M. Wiley
	217	Regret About the Timing of First Sexual Intercourse: The Role of Age and Context <i>Richard Layte</i> , Hannah McGee
	216	Determinants of Water Connection Type and Ownership of Water-Using Appliances in Ireland Joe O'Doherty, <i>Seán Lyons</i> and <i>Richard S.J. Tol</i>

215	Unemployment – Stage or Stigma? Being Unemployed During an Economic Boom <i>Emer Smyth</i>
214	The Value of Lost Load Richard S.J. Tol
213	Adolescents' Educational Attainment and School Experiences in Contemporary Ireland Merike Darmody, Selina McCoy, Emer Smyth
212	Acting Up or Opting Out? Truancy in Irish Secondary Schools <i>Merike Darmody, Emer Smyth</i> and <i>Selina McCoy</i>
211	Where do MNEs Expand Production: Location Choices of the Pharmaceutical Industry in Europe after 1992 <i>Frances P. Ruane</i> , Xiaoheng Zhang
210	Holiday Destinations: Understanding the Travel Choices of Irish Tourists <i>Seán Lyons, Karen Mayor</i> and <i>Richard S.J. Tol</i>
209	The Effectiveness of Competition Policy and the Price- Cost Margin: Evidence from Panel Data Patrick McCloughan, <i>Seán Lyons</i> and William Batt
208	Tax Structure and Female Labour Market Participation: Evidence from Ireland <i>Tim Callan</i> , A. Van Soest, <i>J.R. Walsh</i>
207	Distributional Effects of Public Education Transfers in Seven European Countries <i>Tim Callan,</i> Tim Smeeding and Panos Tsakloglou